Author’s response to reviews

Title: Effectiveness of Trauma Team on Medical Resource Utilization and Quality of Care for Patients with Major Trauma

Authors:

Chih-Jung Wang (poemcage@gmail.com)
Shu-Ting Yen (j_yen33@ymail.com)
Shih-Fang Huang (cycu0621@gmail.com)
Su-Chen Hsu (suchenhsu@gmail.com)
Jeremy Ying (jying@isu.edu.tw)
Yan-Shen Shan (ysshan@mail.ncku.edu.tw)

Version: 2 Date: 21 Jun 2017

Author’s response to reviews:

20170621

Dear Editor:

The following is the response to critique of submitted manuscript: BHSR-D-16-01102R1; Effectiveness of Trauma Team on Medical Resource Utilization and Quality of Care for Patients with Major Trauma. Chih-Jung Wang; Shu-Ting Yen; Shih-Fang Huang, BS; Su-Chen Hsu; Jeremy C Ying, PhD; Yan-Shen Shan

1) In the Consent to publish section, this section is for if patients have given consent to publish their individual details. As this is not applicable to your study, please state "Not applicable".

I have stated not applicable in the Consent to publish section (page 9).
2) Please update your Availability of data and materials section to clarify where the data supporting your findings can be found (including information found in any supplementary files). If you do not wish to share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason. If all the data supporting your findings is contained within the manuscript, please state so in the “Availability of data and materials” section. Please note that identifying/confidential patient data should not be shared.

I have added Availability of data and materials section and explained the finding in the manuscript (page 9).

3) Please update your Competing interests statement to declare all financial and non-financial competing interests. See our editorial policies (http://www.biomedcentral.com/submissions/editorial-policies#competing+interests) for a full explanation of competing interests. If you are unsure whether you or any of your co-authors have a competing interest please contact the editorial office.

I have updated a full explanation of competing interests (page 9).

4) Please use initials for authors in the Authors contributions statement.

I have corrected initials for authors in the section of Author’s contributions (page 9–page 10).

5) Please remove the Copyright Transfer Agreement before resubmitting your manuscript.

The Copyright Transfer Agreement is not included.

Yan-Shen Shan MD,PhD
Distinguished Prof and Director
Department of Clinical Medicine Research and Surgery, NCKUH
Institute of Clinical Medicine, College of Medicine, NCKU

ysshan@mail.ncku.edu.tw